Author’s response to reviews

Title: Polypharmacy in a hospitalized psychiatric population: risk estimation and damage quantification

Authors:
JAIME CARMONA-HUERTA (JAIME.CARMONA@ACADEMICOS.UDG.MX)
SANTIAGO CASTIELLO-DE-OBESO (santiagocdo@gmail.com)
JULISSA RAMIREZ-PALOMINO (julissa.areli.palomino@hotmail.com)
RODRIGO DURAN-GUTIERREZ (med04rodrigo@gmail.com)
DAVID CARDONA-MULLER (david.cardona@academicos.udg.mx)
FERNANDO GROVER-PAEZ (fgroverp@hotmail.com)
PAOLA FERNANDEZ-DORANTES (pa-fdezd@hotmail.com)
RAFAEL MEDINA-DAVALOS (medina_666@hotmail.com)

Version: 3 Date: 30 Jan 2019

Author’s response to reviews:

Corrections

• Regarding the last reviewers comments we corrected the following:

  o Page 11 subtitle “Substance Abuse and comorbidities” paragraph 1 line 9: The legend “table 3” was changed to “table 2” which the correct table related to in the paragraph

  o Page 12 subtitle “Drug Use and Polypharmacy” paragraph 1 line 2: a new decimal was added to the number 7.8 (changed to 7.88) so that it is the same number mentioned in table 3

  o Page 13 subtitle “Drug use and polypharmacy” paragraph 3 line 3: olanzapine 22% was changed to olanzapine 22.9% which is the same number written in table 3

  o Page 13 subtitle “Drug use and polypharmacy” paragraph 3 line 1-3: the first 3 lines of the paragraph were restructured to make it clear that Haloperidol has 2 presentations that when combined add up to 30% of prescription.
Page 13 subtitle “Drug use and polypharmacy” paragraph 4 line 4: the legend “table 2” was changed for “table 3” which is the table that the paragraph refers to.